

GOPEN ACCESS

Citation: Eisenga MF, De Jong MA, Van der Meer P, Leaf DE, Huls G, Nolte IM, et al. (2019) Iron deficiency, elevated erythropoietin, fibroblast growth factor 23, and mortality in the general population of the Netherlands: A cohort study. PLoS Med 16(6): e1002818. https://doi.org/ 10.1371/journal.pmed.1002818

Academic Editor: Maarten W. Taal, Royal Derby Hospital, UNITED KINGDOM

Received: January 4, 2019

Accepted: May 2, 2019

Published: June 6, 2019

Copyright: © 2019 Eisenga et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in this study can be made available by the data manager of the PREVEND study, Ms. Lyanne Kieneker, I.m.kieneker@umcg. nl. Public sharing of individual participant data was not included in the informed consent form of the study, but data can be made available to interested researchers upon reasonable request.

Funding: This work was supported by the Dutch Kidney Foundation (Kolff senior postdoc grant to

RESEARCH ARTICLE

Iron deficiency, elevated erythropoietin, fibroblast growth factor 23, and mortality in the general population of the Netherlands: A cohort study

Michele F. Eisenga¹*, Maarten A. De Jong¹, Peter Van der Meer², David E. Leaf³, Gerwin Huls⁴, Ilja M. Nolte⁵, Carlo A. J. M. Gaillard⁶, Stephan J. L. Bakker¹, Martin H. De Borst¹

 Division of Nephrology, Department of Internal Medicine, University of Groningen, University Medical Center Groningen, Groningen, the Netherlands, 2 Department of Cardiology, University of Groningen, University Medical Center Groningen, Groningen, the Netherlands, 3 Division of Renal Medicine, Brigham and Women's Hospital, Harvard Medical School, Boston, Massachusetts, United States of America,
Division of Hematology, Department of Internal Medicine, University of Groningen, University Medical Center Groningen, Groningen, the Netherlands, 5 Department of Epidemiology, University of Groningen, University Medical Center Groningen, Groningen, the Netherlands, 6 Department of Internal Medicine and Dermatology, University of Utrecht, University Medical Center Utrecht, Utrecht, the Netherlands

* m.f.eisenga@umcg.nl

Abstract

Background

Emerging data in chronic kidney disease (CKD) patients suggest that iron deficiency and higher circulating levels of erythropoietin (EPO) stimulate the expression and concomitant cleavage of the osteocyte-derived, phosphate-regulating hormone fibroblast growth factor 23 (FGF23), a risk factor for premature mortality. To date, clinical implications of iron deficiency and high EPO levels in the general population, and the potential downstream role of FGF23, are unclear. Therefore, we aimed to determine the associations between iron deficiency and higher EPO levels with mortality, and the potential mediating role of FGF23, in a cohort of community-dwelling subjects.

Methods and findings

We analyzed 6,544 community-dwelling subjects (age 53 ± 12 years; 50% males) who participated in the Prevention of Renal and Vascular End-Stage Disease (PREVEND) study—a prospective population-based cohort study, of which we used the second survey (2001– 2003)—and follow-up was performed for a median of 8 years. We measured circulating parameters of iron status, EPO levels, and plasma total FGF23 levels. Our primary outcome was all-cause mortality. In multivariable linear regression analyses, ferritin ($\beta = -0.43$), transferrin saturation (TSAT) ($\beta = -0.17$), hepcidin ($\beta = -0.36$), soluble transferrin receptor (sTfR; $\beta = 0.33$), and EPO ($\beta = 0.28$) were associated with FGF23 level, independent of potential confounders. During median (interquartile range [IQR]) follow-up of 8.2 (7.7–8.8) years, 379 (6%) subjects died. In multivariable Cox regression analyses, lower levels of MHDB, Grant Number 170KG18). The funder did not influence the design, development, analysis, or interpretation of the current study, nor did it contribute to the current manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: MFE received speaking fees from Vifor Pharma. PvdM received personal fees and research grants from Vifor Pharma. CAJMG received speaking fees and research funding from Vifor Pharma, served on advisory boards for Vifor, Amgen, Roche, and Pharmacosmos, and was a member of the FIND CKD steering committee. SJLB is a member of the Dutch Health Council and received research grants from Astellas Pharma, Chiesi, Nedmag, LabCorp. Vifor Pharma, FrieslandCampina, TiFN, DSM, Toegepaste en Technische Wetenschappen, Nutricia, and the Dutch Kidney Foundation. MHdB has served as a consultant or received honoraria (to employer) from Amgen, Bayer, Sanofi Genzyme, Kyowa Kirin Pharma, and Vifor Fresenius Medical Care Renal Pharma.

Abbreviations: 25D, 25-hydroxyvitamin D; ACE, angiotensin converting enzyme; BMI, body mass index; CI, confidence interval; CKD, chronic kidney disease; CKD-EPI, Chronic Kidney Disease Epidemiology Collaboration; CV, coefficient of variation; eGFR, estimated glomerular filtration rate; ELISA, enzyme-linked immunosorbent assay; EPO, erythropoietin; ESA, EPO-stimulating agent; Fam20C, family with sequence similarity 20, member C; FGF23, fibroblast growth factor 23; GALNT3, N-acetylgalactosaminyltransferase 3; HR, hazard ratio; hs-CRP, high-sensitivity C-reactive protein; iFGF23, intact FGF23; IQR, interquartile range; MCV, mean corpuscular volume; PREVEND, Prevention of Renal and Vascular End-Stage Disease; PTH, parathyroid hormone; SD, standard deviation; SI, system of units; sTfR, soluble transferrin receptor; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology; TSAT, transferrin saturation.

TSAT (hazard ratio [HR] per 1 standard deviation [SD], 0.84; 95% confidence interval [CI], 0.75–0.95; P = 0.004) and higher levels of sTfR (HR, 1.15; 95% CI 1.03–1.28; P = 0.01), EPO (HR, 1.17; 95% CI 1.05–1.29; P = 0.004), and FGF23 (HR, 1.20; 95% CI 1.10–1.32; P < 0.001) were each significantly associated with an increased risk of death, independent of potential confounders. Adjustment for FGF23 levels markedly attenuated the associations of TSAT (HR, 0.89; 95% CI 0.78–1.01; P = 0.06), sTfR (HR, 1.08; 95% CI 0.96–1.20; P = 0.19), and EPO (HR, 1.10; 95% CI 0.99–1.22; P = 0.08) with mortality. FGF23 remained associated with mortality (HR, 1.15; 95% CI 1.04–1.27; P = 0.008) after adjustment for TSAT, sTfR, and EPO levels. Mediation analysis indicated that FGF23 explained 31% of the association between TSAT and mortality; similarly, FGF23 explained 32% of the association between sTfR and mortality and 48% of the association between EPO and mortality (indirect effect P < 0.05 for all analyses). The main limitations of this study were the observational study design and the absence of data on intact FGF23 (iFGF23), precluding us from discerning whether the current results are attributable to an increase in iFGF23 or in C-terminal FGF23 fragments.

Conclusions and relevance

In this study, we found that functional iron deficiency and higher EPO levels were each associated with an increased risk of death in the general population. Our findings suggest that FGF23 could be involved in the association between functional iron deficiency and increased EPO levels and death. Investigation of strategies aimed at correcting iron deficiency and reducing FGF23 levels is warranted.

Author summary

Why was this study done?

- Iron deficiency, one of the most common nutritional disorders worldwide, is known to be associated with increased risk of death in the general population, although a clear mechanism has not been identified.
- In patient populations, it has been shown that higher levels of erythropoietin (EPO) are associated with adverse outcomes, but its significance in the general population has not been elucidated yet.
- Emerging data from the chronic kidney disease (CKD) field suggest that both iron deficiency and EPO lead to an up-regulated expression and concomitant cleavage of a phosphate-regulating hormone, fibroblast growth factor 23 (FGF23), another risk factor for mortality.

What did the researchers do and find?

• We used the Prevention of Renal and Vascular End-Stage Disease (PREVEND) study, a prospective population-based cohort study from 2001 to 2003 involving 6,588 subjects and for which follow-up has been performed for a median of 8 years.

- We found that iron status parameters and EPO were the strongest determinants of FGF23 in the general population, even more than such well-described determinants as phosphate, kidney function, and calcium levels.
- Functional iron deficiency and EPO were associated with an increased risk of death in the general population, and FGF23 explained a substantial part of these associations.

What do these findings mean?

- The current findings suggest that FGF23 is important in the pathophysiology of irondeficiency-and EPO-mediated mortality in the population.
- This study provides the rationale to assess the potential benefits of iron supplementation to reduce mortality in the general population, possibly by lowering FGF23 levels.

Introduction

Iron deficiency is one of the most common nutritional disorders worldwide [1]. In addition to its effect on quality of life and functional capacity, iron deficiency has previously been associated with an increased risk of all-cause and cardiovascular mortality in the general population [2]. However, the underlying mechanisms linking iron deficiency with mortality in this setting remain unknown. Emerging data from disease populations such as chronic kidney disease (CKD) patients suggest that iron deficiency stimulates the expression and concomitant cleavage of the osteocyte-derived, phosphate-regulating hormone fibroblast growth factor 23 (FGF23). Elevated levels of FGF23, in turn, have been shown to be associated with an increased risk of mortality in both community-dwelling individuals [3] and across stages of CKD [4–6].

Similarly, higher circulating endogenous erythropoietin (EPO) levels have been associated with an increased all-cause and cardiovascular mortality risk in various disease populations, including chronic heart failure patients, kidney transplant recipients, and elderly individuals [7–10]. In addition, higher doses of exogenous EPO increase the risk of cardiovascular events in CKD patients [11–13]. To date, it is unknown whether EPO levels are associated with adverse outcomes in the general population. Similar to iron deficiency, recent studies from our group and others have established that both endogenous and exogenous EPO may influence FGF23 production and metabolism in CKD patients [14–16].

Currently, the clinical implications of iron deficiency and high EPO levels in the general population, as well as the potential downstream role for FGF23, are unclear. Hence, in the current study, we investigated whether iron deficiency and EPO influence FGF23 levels, whether iron deficiency and elevated EPO levels are associated with an increased risk of death, and whether such associations could be explained by variation in FGF23 levels.

Methods

Study population

Details from the Prevention of Renal and Vascular End-Stage Disease (PREVEND) study have been described previously [17]. This study was planned following prior results indicating that iron deficiency and high EPO levels are associated with higher FGF23 levels and subsequently

contribute to an increased mortality risk in kidney transplant recipients [16,18]. Between January and September 2018, we addressed the hypothesis that similar relationships would exist in the general population. For this study, we did not have a prespecified analysis plan, but we performed hypothesis-driven analyses in which no data-driven changes have taken place. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (S1 Checklist). In brief, from 1997 to 1998, all inhabitants of the city of Groningen, the Netherlands, who were 28 to 75 years old received a questionnaire on demographics, disease history, smoking habits, use of medication, and a vial to collect an early morning urinary sample (n = 85,421). Of these individuals, 40,856 (47.8%) responded. After exclusion of subjects with insulin-dependent diabetes mellitus and pregnant women, subjects with a urinary albumin concentration ≥ 10 mg/L (n = 6,000) and a randomly selected control group with a urinary albumin excretion <10 mg/L (n = 2,592) completed the screening protocol and, as such, formed the baseline PREVEND cohort (n = 8,592). For the current analyses, we used data from the second survey, which took place between 2001 and 2003 (n = 6,894), since for this visit, blood samples were also available. We excluded 436 subjects due to missing data on FGF23, resulting in 6,544 subjects (visually depicted in a flowchart in S1 Fig). The PREVEND study protocol was approved by the institutional medical review board and was in accordance with the Declaration of Helsinki. Written informed consent was obtained from all subjects.

Exposures and outcomes

Fasting blood samples were drawn in the morning from all subjects from April 24, 2001, to December 3, 2003. All hematologic measurements were measured in fresh venous blood. Aliquots of these samples were stored immediately at -80°C until further analysis. Serum creatinine was measured using an enzymatic method on a Roche Modular analyzer (Roche Diagnostics, Mannheim, Germany). For predicting the estimated glomerular filtration rate (eGFR), the Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI) was applied [19]. Serum iron was measured using a colorimetric assay, ferritin using immunoassay, and transferrin using an immunoturbidimetric assay (all Roche Diagnostics). Transferrin saturation (TSAT, %) was calculated as 100 × serum iron (μ mol/L) \div 25 × transferrin (g/L) [20]. Serum EPO was measured using an immunoassay based on chemiluminescence (Immulite EPO assay, DPC, Los Angeles, CA). Serum hepcidin was measured with a competitive enzyme-linked immunosorbent assay (ELISA), as described elsewhere with intra- and interassay coefficients of variation (CVs) of 8.6% and 16.2%, respectively [21]. Soluble transferrin receptor (sTfR) was measured using an automated homogenous immunoturbidimetric assay with intra- and interassay CVs <2% and <5% [22]. Total FGF23 levels were measured in plasma EDTA samples with a human FGF23 ELISA (Quidel Corp., San Diego, CA) directed against 2 different epitopes within the C-terminal part of the FGF23 molecule. Hence, the assay measures both the intact hormone and the C-terminal fragments and, as such, measures total FGF23 levels. In our hands, this ELISA had intra- and interassay CVs of <5% and <16% in blinded replicated samples, respectively [23]. We assessed prospective associations of iron status parameters, EPO, hepcidin, and FGF23 levels with death. In the PREVEND cohort, data on mortality were received through the municipal register, and follow-up was available until January 1, 2011.

Statistical analyses

Data were analyzed using IBM SPSS software, version 23.0 (SPSS, Chicago, IL), R version 3.2.3 (Vienna, Austria), and STATA 14.1 (STATA Corp., College Station, TX). Baseline characteristics are described as means with standard deviation (SD) for normally distributed variables, as medians with interquartile range (IQR) for skewed variables, or as numbers with corresponding

percentages for categorical variables. Differences in baseline characteristics across tertiles of FGF23 were tested with a one-way ANOVA for normally distributed variables, Kruskal-Wallis test for skewed variables, and a chi-squared test for categorical variables. Hereafter, we assessed the relationship between iron status parameters and EPO as determinants of FGF23 by calculating Spearman correlation coefficients. We subsequently performed multivariable analyses adjusted for age, sex, eGFR, urinary albumin excretion, systolic blood pressure, alcohol use (5 categories), smoking status (never, former, or current smoker), hemoglobin, mean corpuscular volume (MCV), and plasma/serum levels of high-sensitivity C-reactive protein (hs-CRP), phosphate, calcium, 25-hydroxyvitamin D (25D), and parathyroid hormone (PTH). In all analyses, skewed variables, i.e., FGF23, ferritin, hepcidin, sTfR, EPO, urinary albumin excretion, PTH, and hs-CRP, were naturally log transformed prior to inclusion. Second, we performed stepwise backward multivariable linear regression analyses in which we determined whether iron status parameters and EPO remained major determinants of FGF23 levels. Of note, because the different iron status parameters are highly correlated with each other (S1 Table), we included all iron status parameters individually in multivariable linear regression analyses. To visualize the crosssectional associations between the different iron status parameters-EPO and FGF23-plots were generated using locally weighted scatterplot smoothing.

We subsequently assessed the associations of iron status parameters and EPO levels with all-cause mortality using Cox proportional hazard regression analyses. Assumption of proportionality in Cox regression analyses was checked using Schoenfeld residuals plots. We constructed multivariable models adjusted for age and sex (model 1), and additionally for eGFR, urinary albumin excretion, body mass index (BMI), systolic blood pressure, hs-CRP, presence of diabetes, smoking, alcohol use, and use of antihypertensives (model 2). Subsequently, we adjusted for FGF23 (model 3). Similarly, we assessed whether FGF23 levels were associated with mortality, independent of potential confounders, including adjustment for iron status parameters and EPO in the final models. Finally, we performed mediation analyses with the methods described by Preacher and Hayes, which is based on logistic regression. These analyses allow for testing significance and magnitude of mediation [24,25]. Overall, 3.4% of demographic data were missing and were imputed using regressive switching [26]. Five datasets were multiply imputed, and results were pooled according to Rubin's rules [27]. In all analyses, a two-sided significance level <0.05 was considered significant.

Results

Baseline characteristics

We included 6,544 community-dwelling subjects (age 53 ± 12 years; 50% males) with a median (IQR) FGF23 level of 70 (57–87) RU/mL. Mean hemoglobin concentration was 13.7 ± 1.2 g/dL; median ferritin concentration was 96 (47–172) µg/L; mean TSAT was $25.0 \pm 9.5\%$; median sTfR level was 2.5 (2.1–3.0) mg/L; median hepcidin level was 8.5 (4.6–13.8) ng/L; and median EPO level was 7.8 (5.9–10.3) IU/L. Iron deficiency based on ferritin levels (i.e., ferritin < 15 µg/L for women and < 30 µg/L for men) was present in 448 (7%) individuals [28]. Iron deficiency based on low TSAT levels (i.e., TSAT < 20%) was present in 1,806 (28%) individuals [2]. EPO-stimulating agents (ESAs) and oral or intravenous iron were not used by any of the subjects. Across tertiles of FGF23, we observed inverse associations with ferritin, TSAT, and hepcidin and a positive association with sTfR and EPO levels. Additional demographic, clinical, and laboratory parameters are depicted in Table 1.

Iron parameters, EPO, and FGF23

In univariate analyses, higher FGF23 levels correlated with lower levels of ferritin ($\rho = -0.21$, P < 0.001), TSAT ($\rho = -0.21$, P < 0.001), sTfR ($\rho = 0.20$, P < 0.001), and hepcidin ($\rho = -0.18$,

		Tertiles of total FGF23 (RU/mL)			
Baseline characteristic	All patients (<i>n</i> = 6,544)	T1 (<i>n</i> = 2,177)	T2 (<i>n</i> = 2,186)	T3 (<i>n</i> = 2,181)	P value
		[20.7-60.6]	[60.7-80.2]	[80.3-3,494.6]	-
Age (y)	53 ± 12	51 ± 12	53 ± 12	55 ± 12	< 0.001
Male sex (n, %)	3,251 (50)	1,193 (55)	1,152 (53)	906 (42)	< 0.001
BMI, kg/m ²	26.7 ± 4.4	26.1 ± 3.8	26.6 ± 4.3	27.5 ± 4.8	< 0.001
Alcohol use					< 0.001
No alcohol use (<i>n</i> , %)	1,722 (26)	482 (22)	534 (24)	706 (32)	
1–4 units/mo (<i>n</i> , %)	1,093 (17)	386 (18)	360 (17)	347 (16)	
2–7 units/wk (<i>n</i> , %)	2042 (31)	730 (34)	675 (31)	637 (29)	
>1-3 units/d (<i>n</i> , %)	1,416 (22)	498 (23)	515 (24)	403 (19)	
>3 units/d (<i>n</i> , %)	271 (4)	81 (4)	102 (5)	88 (4)	
Smoking status					< 0.001
Never smoker (<i>n</i> , %)	1,967 (30)	756 (35)	647 (30)	564 (26)	
Former smoker (<i>n</i> , %)	2,748 (42)	957 (44)	945 (43)	846 (39)	
Current smoker (<i>n</i> , %)	1,829 (28)	464 (21)	594 (27)	771 (35)	
Diabetes mellitus (<i>n</i> , %)	403 (6)	95 (4)	118 (5)	190 (9)	< 0.001
Systolic blood pressure (mmHg)	126 ± 19	125 ± 18	126 ± 19	128 ± 20	0.001
Diastolic blood pressure (mmHg)	73 ± 9	73 ± 9	73 ± 9	74 ± 9	0.90
Laboratory measurements					
Iron status					
Ferritin (µg/L)	96 (47-172)	110 (61–190)	102 (53–178)	74 (28–149)	< 0.001
TSAT (%)	25.0 ± 9.5	26.8 ± 9.3	25.9 ± 9.1	22.5 ± 9.5	< 0.001
sTfR (mg/L)	2.5 (2.1-3.0)	2.4 (2.1-2.8)	2.4 (2.1-2.9)	2.7 (2.2-3.3)	< 0.001
Hepcidin (ng/mL)	8.5 (4.6–13.8)	9.4 (5.5–14.5)	8.9 (5.2–13.9)	7.1 (2.9–12.8)	< 0.001
EPO (IU/L)	7.8 (5.9–10.3)	7.2 (5.6–9.3)	7.7 (5.9–9.9)	8.6 (6.4–11.8)	< 0.001
Hemoglobin (g/dL)	13.7 ± 1.2	13.8 ± 1.1	13.8 ± 1.1	13.6 ± 1.4	< 0.001
MCV (fL)	90 ± 5	91 ± 4	91 ± 4	90 ± 6	< 0.001
Total cholesterol (mg/dL)	209.6 ± 40.4	208.5 ± 40.2	209.8 ± 40.4	210.4 ± 40.6	0.17
Glucose (mg/dL)	91 ± 21	89 ± 18	90 ± 20	93 ± 24	< 0.001
Phosphate (mg/dL)	3.13 ± 0.87	3.07 ± 0.89	3.10 ± 0.74	3.21 ± 0.95	< 0.001
Calcium (mg/dL)	9.23 ± 0.46	9.19 ± 0.44	9.24 ± 0.47	9.26 ± 0.47	0.18
PTH (pg/mL)	46.0 (38.4–55.3)	44.9 (37.5–53.5)	45.5 (38.4-54.4)	47.5 (39.2–57.6)	< 0.001
25 (OH) vitamin D (ng/mL)	22.8 ± 10.0	23.4 ± 10.2	23.0 ± 9.9	21.9 ± 10.0	< 0.001
eGFR (ml/min/1.73 m ²)	91.8 ± 17.2	98.5 ± 14.6	92.4 ± 15.4	85.6 ± 18.7	< 0.001
Creatinine (mg/dL)	0.96 ± 0.24	0.93 ± 0.15	0.96 ± 0.16	0.99 ± 0.35	< 0.001
Urinary albumin excretion (mg/24 h)	8.7 (6.1–16.0)	8.4 (6.0-14.6)	8.5 (6.1-14.7)	9.5 (6.2-20.3)	< 0.001
hs-CRP (mg/L)	1.4 (0.6–3.1)	1.1 (0.5–2.5)	1.3 (0.6–2.9)	1.7 (0.8–3.9)	< 0.001
Medication use					
Antihypertensives* (n, %)	1,299 (20)	269 (12)	397 (18)	633 (29)	< 0.001
ACE inhibitors or AII antagonists $(n, \%)$	580 (9)	126 (6)	198 (9)	256 (12)	< 0.001

Table 1. Baseline characteristics of 6,544 community-dwelling subjects according to tertiles of total FGF23 levels.

*Includes ACE or AII antagonists; *P* for trend was calculated with linear regression for continuous variables and with chi-squared test for dichotomous and categorical variables. SI conversion factors: to convert hepcidin from ng/mL to nmol/L, divide by 2.789; to convert hemoglobin from g/dL to mmol/L, multiply by 0.6206; to convert total cholesterol from mg/dL to mmol/L, multiply by 0.0259; to convert glucose from mg/dL to mmol/L, multiply by 0.0555; to convert phosphate from mg/dL to mmol/L, multiply by 0.3229; to convert calcium from mg/dL to mmol/L, multiply by 0.2495; to convert PTH from pg/mL to pmol/L, multiply by 0.105; to convert vitamin D from ng/mL to nmol/L, multiply by 2.496; to convert creatinine from mg/dL to µmol/L, multiply by 88.42.

Abbreviations: ACE, angiotensin converting enzyme; BMI, body mass index; CRP, C-reactive protein; eGFR, estimated glomerular filtration rate; EPO, erythropoietin; FGF23, fibroblast growth factor 23; hs-CRP, high-sensitivity C-reactive protein; MCV, mean corpuscular volume; PTH, parathyroid hormone; SI, system of units; sTfR, soluble transferrin receptor; TSAT, transferrin saturation.

https://doi.org/10.1371/journal.pmed.1002818.t001



Fig 1. Cross-sectional associations of iron status parameters and EPO with FGF23. Plots were generated with use of locally weighted scatterplot smoothing and show the association of FGF23 with ferritin (Panel A), TSAT (Panel B), sTfR (Panel C), hepcidin (Panel D), and EPO (Panel E). Lines and band represent means and 95% CIs, respectively. CI, confidence interval; FGF23, ferritin, sTfR, hepcidin, and EPO were naturally log transformed. EPO, erythropoietin; FGF23, fibroblast growth factor 23; sTfR, soluble transferrin receptor; TSAT, transferrin saturation.

https://doi.org/10.1371/journal.pmed.1002818.g001

P < 0.001) and with higher levels of serum EPO ($\rho = 0.19, P < 0.001$) (Fig 1A–1E). After adjustment for age, sex, eGFR, urinary albumin excretion, systolic blood pressure, alcohol use, smoking status, hemoglobin, MCV, hs-CRP, phosphate, calcium, 25D, and PTH, all iron parameters—including ferritin ($\beta = -0.43, P < 0.001$), TSAT ($\beta = -0.17, P < 0.001$), sTfR ($\beta = 0.33, P < 0.001$), and hepcidin ($\beta = -0.36, P < 0.001$)—remained strongly and independently associated with FGF23 levels. Similarly, EPO ($\beta = 0.28, P < 0.001$) remained independently

associated with FGF23. In stepwise backward linear regression analyses, ferritin ($\beta = -0.38$, P < 0.001), sTfR ($\beta = 0.27$, P < 0.001), hepcidin ($\beta = -0.31$, P < 0.001), and EPO ($\beta = 0.21$, P < 0.001) levels were identified as the strongest determinants of FGF23 levels, with higher standardized regression coefficients than more established determinants of FGF23, including eGFR ($\beta = -0.20$, P < 0.001), phosphate ($\beta = 0.13$, P < 0.001), and calcium ($\beta = 0.17$, P < 0.001) (S2 Table).

Iron status and mortality

During a median (IQR) follow-up of 8.2 (7.7–8.8) years, 379 (6%) subjects died. After 1, 5, and 8 years of follow-up, 6,437, 6,040, and 3,993 subjects, respectively, were still followed up in the study. We assessed the associations of the individual iron status parameters, EPO and FGF23, with mortality (Table 2). After adjustment for age and sex, neither ferritin nor hepcidin was associated with risk of death (hazard ratio [HR] per 1 SD higher ln[ferritin], 0.95; 95% confidence interval [CI], 0.84–1.07; P = 0.38; HR per 1 SD higher ln[hepcidin], 0.99; 95% CI 0.88–1.11; P = 0.80, respectively). In contrast, a lower TSAT was strongly associated with an increased risk of death (age- and sex-adjusted HR per 1 SD higher TSAT, 0.79; 95% CI 0.70–0.88; P < 0.001). After further adjustment for eGFR, urinary albumin excretion, BMI, systolic blood pressure, hs-CRP, presence of diabetes, smoking, alcohol use, and use of antihypertensives (model 2), the association between TSAT and mortality persisted (HR, 0.84; 95% CI 0.75–0.95; P = 0.004). However, subsequent adjustment for FGF23 attenuated the association, such that TSAT was no longer significantly associated with death (HR, 0.89; 95% CI 0.78–1.01; P = 0.06).

Similarly, a higher plasma sTfR level, indicating functional iron deficiency, was associated with a higher risk of mortality in a model adjusted for age and sex (HR per 1 SD higher ln [sTfR], 1.17; 95% CI 1.06–1.30; P = 0.004). In multivariable analyses (model 2), the association between sTfR and mortality remained significant (HR, 1.15; 95% CI 1.03–1.28; P = 0.01). However, adjustment for FGF23 strongly attenuated the association, rendering the association between sTfR and mortality nonsignificant (HR, 1.08; 95% CI 0.96–1.20; P = 0.19).

In mediation analyses, we analyzed whether the significant associations between iron status parameters (i.e., TSAT and sTfR) and mortality were mediated by FGF23 (Table 3). FGF23

Biomarker	Model 1	Model 2	Model 3
FGF23	1.29 (1.20–1.34)	1.20 (1.10–1.32)	1.15 (1.04–1.27)
EPO	1.22 (1.10–1.34)	1.17 (1.05–1.29)	1.10 (0.99–1.22)
TSAT	0.79 (0.70-0.88)	0.84 (0.75-0.95)	0.89 (0.78-1.01)
sTfR	1.17 (1.06–1.30)	1.15 (1.03–1.28)	1.08 (0.96-1.20)
Ferritin	0.95 (0.84–1.07)	0.91 (0.80-1.03)	1.02 (0.90-1.16)
Hepcidin	0.99 (0.88–1.11)	0.95 (0.85-1.07)	1.03 (0.91-1.16)

Table 2. HRs (and 95% CIs) for death according to FGF23, iron status parameters, and EPO.

The table shows the different HRs for death according to the different natural log-transformed (except TSAT) parameters standardized to 1 SD. Model 1 is adjusted for age and sex; model 2 is adjusted for eGFR, urinary albumin excretion, BMI, systolic blood pressure, hs-CRP, presence of diabetes, smoking, alcohol use, and use of antihypertensives; for FGF23, extra adjustments for calcium, phosphate, 25 vitamin D, and PTH. Model 3 has been adjusted for FGF23 in the EPO and iron analyses, whereas it has been adjusted for EPO, TSAT, and sTfR in the FGF23 analysis.

Abbreviations: CI, confidence interval; eGFR, estimated glomerular filtration rate; EPO, erythropoietin; FGF23, fibroblast growth factor 23; HR, hazard ratio; hs-CRP, high-sensitivity C-reactive protein; PTH, parathyroid hormone; SD, standard deviation; sTfR, soluble transferrin receptor; TSAT, transferrin saturation.

https://doi.org/10.1371/journal.pmed.1002818.t002

Independent variable	Potential mediator	Outcome	Effect (path)*	Multivariable model**	
				Coefficient (95% CI, bc)***	Proportion mediated†
TSAT	FGF23	Mortality	Indirect effect (ab path)	-0.030 (-0.050 to -0.012)	31%
			Total effect ($ab + c'$ path)	-0.098 (-0.174 to -0.026)	
sTfR	FGF23	Mortality	Indirect effect (ab path)	0.039 (0.009 to 0.072)	32%
			Total effect ($ab + c'$ path)	0.123 (0.040 to 0.205)	
EPO	FGF23	Mortality	Indirect effect (ab path)	0.034 (0.012 to 0.060)	48%
			Total effect ($ab + c'$ path)	0.071 (0.008 to 0.139)	
EPO	TSAT	FGF23	Indirect effect (ab path)	0.037 (0.028 to 0.047)	12%
			Total effect ($ab + c'$ path)	0.318 (0.275 to 0.365)	
EPO	sTfR	FGF23	Indirect effect (ab path)	0.108 (0.086 to 0.133)	33%
			Total effect ($ab + c'$ path)	0.326 (0.277 to 0.379)	

Table 3. Mediation analyses of FGF23 on the	ssociation between TSAT, sTfR, EPO	, and mortality in the general population
---	------------------------------------	---

*The coefficients of the indirect *ab* path and the total *ab* + *c*' path are standardized for the SDs of TSAT, sTfR, EPO, FGF23, and all-cause mortality.

** All coefficients are adjusted for age, sex, eGFR, urinary albumin excretion, BMI, systolic blood pressure, hs-CRP, presence of diabetes, smoking, alcohol use, and use of antihypertensives.

***95% CIs for the indirect and total effects were bias-corrected CIs after running 2,000 bootstrap samples.

†The size of the significant mediated effect is calculated as the standardized indirect effect divided by the standardized total effect multiplied by 100. Abbreviations: bc, bias corrected; BMI, body mass index; CI, confidence interval; EPO, erythropoietin; eGFR, estimated glomerular filtration rate; FGF23, fibroblast growth factor 23; hs-CRP, high-sensitivity C-reactive protein; SD, standard deviation; sTfR, soluble transferrin receptor; TSAT, transferrin saturation.

https://doi.org/10.1371/journal.pmed.1002818.t003

was identified as a significant mediator (indirect effect P < 0.05; 31% of the association between TSAT and mortality was explained by FGF23 and 32% of the association between sTfR and mortality).

EPO and mortality

In age- and sex-adjusted analyses, higher serum EPO levels were associated with an increased risk of death (HR per 1 SD higher ln[EPO], 1.22; 95% CI 1.10–1.34; P < 0.001, Table 2). In multivariable analyses (model 2), the association between EPO and mortality persisted (HR, 1.17; 95% CI 1.05–1.29; P = 0.004). However, adjustment for FGF23 abrogated the association between EPO and mortality, rendering the association nonsignificant (HR, 1.10; 95% CI 0.99–1.22; P = 0.08).

In mediation analyses, we analyzed whether the significant association between EPO and mortality was mediated by FGF23 (Table 3). FGF23 was identified as a significant mediator (indirect effect P < 0.05; 48% of the association between EPO and mortality was explained by FGF23). Because functional iron deficiency often occurs in states of EPO-mediated erythropoiesis, we also analyzed whether the positive association between EPO and FGF23 might be, at least in part, mediated by TSAT and sTfR. Both parameters were found to be significant mediators in the positive association between EPO and FGF23 (indirect effect P < 0.05; 12% by TSAT and 33% by sTfR, independent of potential confounders, Table 3).

FGF23 and mortality

In age- and sex-adjusted analyses, higher plasma FGF23 levels were strongly associated with an increased risk of death (HR per 1 SD higher ln[FGF23], 1.29; 95% CI 1.20–1.34; P < 0.001, Table 2). In multivariable analyses (model 2) with inclusion of bone mineral parameters (i.e., calcium, phosphate, 25D, and PTH), the association between FGF23 and mortality remained

(HR, 1.20; 95% CI 1.10–1.32; P < 0.001, Table 2). Further adjustment for TSAT, sTfR, and EPO did not materially change the association between FGF23 and mortality (HR, 1.15; 95% CI 1.04–1.27; P = 0.008).

Discussion

In this study, we found that markers of iron deficiency, especially lower iron availability (i.e., lower TSAT and higher sTfR), as well as elevated serum EPO, were associated with an increased risk of mortality in community-dwelling individuals. Notably, we identified FGF23 as a potential mediator of iron-deficiency-and EPO-related mortality. Iron deficiency and elevated levels of EPO were major determinants of FGF23 levels, to a greater extent than more established determinants such as renal function and serum calcium, PTH, and phosphate. FGF23 in itself was strongly associated with mortality independent of adjustment for iron status parameters and EPO. Our findings suggest that FGF23 could be involved in the pathophysiology of iron-deficiency-and EPO-mediated mortality in the population.

We first addressed the relationship between iron deficiency, measured by 4 different parameters, and all-cause mortality in the general population. Our results are consistent with a previous study identifying low TSAT as a predictor of mortality in the general population [2]. Of note, the prospective associations with mortality differed among the various iron parameters. While TSAT and sTfR were strongly associated with mortality, ferritin and hepcidin were not. This discrepancy is most likely explained by the fact that these markers reflect different aspects of iron metabolism. Serum ferritin is a surrogate for body iron stores, but as an acutephase reactant, it is also strongly up-regulated by inflammation, malignancy, and alcohol intake. Hepcidin is also an acute-phase reactant and is highly correlated with serum ferritin. In contrast, TSAT is more a marker of iron availability for erythropoiesis. Elevated levels of sTfR reflect an increased tissue iron demand, but not body iron stores, and are less affected by concomitant chronic disease and inflammation [29]. In the setting of increased metabolic requirements for iron, transferrin receptors are overexpressed on erythroid precursors in the bone marrow and are shed, resulting in increased sTfR levels in the circulation. Hence, an increased sTfR level reflects both erythroid activity and functional iron deficiency. Because functional iron deficiency occurs in patients with significant EPO-mediated erythropoiesis or as a response to treatment with ESAs [30], we also aimed to assess the association between endogenous EPO levels, as a reflection of tissue hypoxia, and mortality. Prior studies conducted in various populations, including in elderly individuals, kidney transplant recipients, and in patients with chronic heart failure, found that higher EPO levels are associated with an increased risk of death, even independent of hemoglobin levels [8-10]. Furthermore, large randomized trials in chronic heart failure and CKD patients striving for stronger correction of anemia with ESAs were associated with an increased risk of mortality [12,13,31,32]. In the current study, we identified for the first time a strong and independent association between higher serum EPO levels and increased mortality in community-dwelling individuals.

The associations we observed between functional iron deficiency, high EPO levels, and mortality led us to explore FGF23 as a potential downstream factor mediating these associations, given accumulating evidence supporting a direct relationship between iron status, EPO, and FGF23 metabolism [16,18,33–35]. Recently, our group and others demonstrated that iron deficiency is a strong determinant of total FGF23 levels in CKD and kidney transplant recipients [18,36]. Mechanistically, it has recently been shown that iron deficiency stabilizes hypoxia-inducible factor 1-alpha, which in turn up-regulates furin, promoting cleavage of the intact FGF23 (iFGF23) molecule into C-terminal FGF23 fragments [34,37,38]. Furthermore, EPO-induced up-regulation of FGF23 production has been demonstrated in CKD patients,

kidney transplant recipients, and in animal models in which circulating EPO levels are elevated due to either endogenous or exogenous sources [14,33]. Currently, the exact mechanism by which EPO increases bone and bone marrow FGF23 transcription and FGF23 post-translational cleavage is unknown; however, Rabadi and colleagues observed in bled mice decreased polypeptide N-acetylgalactosaminyltransferase 3 (GalNT3) bone marrow mRNA expression, which protects iFGF23 from proteolysis by furin, allowing increased FGF23 cleavage [14]. Our group, together with collaborators, recently also showed that in mice with chronically elevated endogenous EPO levels, a decreased GalNT3 bone marrow mRNA expression was present, without differences in family with sequence similarity 20, member C (Fam20C) or furin expression [16]. In the current study, multivariable cross-sectional analyses also demonstrated strong associations of iron parameters and EPO with FGF23, independent of more established determinants of FGF23, including serum phosphate and eGFR.

We subsequently found that the associations between functional iron deficiency-reflected by low TSAT or high sTfR levels—and mortality were mediated by FGF23. Moreover, EPOrelated mortality was also for a considerable part explained by variation in FGF23 levels. Of interest, the positive association between EPO and FGF23 was in part mediated by functional iron deficiency. These findings support our hypothesis that FGF23 is closely related to erythropoiesis and that up-regulation of FGF23 induced by iron deficiency or high EPO levels may subsequently lead to a higher mortality risk. The association between FGF23 and death has been previously demonstrated in different patient populations, including CKD, renal transplant recipients, acute kidney injury, chronic heart failure, and in the general population [3,4,39,40]. The downstream consequences of elevated levels of FGF23 have not been fully elucidated yet. Many reports have revealed that iFGF23 has biologic activity through binding to several FGF23 receptors, including FGFR1, FGFR2, and FGFR4. Besides the classic functions of iFGF23 in regulating renal phosphate handling and vitamin D metabolism, recent studies have demonstrated several "off-target" effects of iFGF23. Preclinical studies demonstrated that FGF23 can induce left ventricular hypertrophy by binding to FGF23 receptor 4 in cardiac myocytes, and promote endothelial dysfunction [41,42]. Furthermore, FGF23 stimulates fibrosis in the kidneys [43], exerts proinflammatory effects by up-regulation of interleukin-6 production [44], and impairs immune function [45]. Finally, it is known that abnormal FGF signaling can promote tumor development by stimulating cancer cell proliferation, invasion, and survival and by supporting tumor angiogenesis [46,47]. Feng and colleagues showed that exogenous FGF23 promotes prostate cancer proliferation, invasion, and independent growth in vitro, whereas FGF23 knockdown in model systems decreased tumor progression both in vitro and in vivo [48]. Given the strong and independent association of FGF23 with mortality and the emerging pathophysiological implications of FGF23, it seems important to unravel major determinants of FGF23 in order to be able to reduce FGF23 levels. In the current study, we have shown that, in the general population, iron deficiency is a major determinant of FGF23 levels, implicating that iron deficiency could potentially be an easily modifiable driver of high FGF23 levels. The potential benefits of iron supplementation to reduce mortality in the general population remain to be addressed in prospective studies.

Our study has several strengths as well as limitations. Major strengths include the availability of FGF23 along with multiple iron status parameters (including hepcidin and sTfR) and EPO in a large population-based cohort. On the other hand, we were unable to measure iFGF23 levels, because samples were not stored with protease inhibitors, and iFGF23 has been shown to be susceptible to degradation with long-term storage [49]. This precludes us from discerning whether the elevated levels of total FGF23 that we observed are attributable to increased circulating levels of intact, biologically active FGF23 or due to increased levels of Cterminal fragments, which are not biologically active. The latter could be due to increased production of FGF23 matched by a concomitant increase in FGF23 cleavage. This pattern has been observed in previous studies in various disease populations, and we speculate that the same pattern might occur in the general population as well. Furthermore, due to the observational design, residual confounding might have biased the results despite the substantial number of potentially confounding factors for which we adjusted. For example, it is known that TSAT is also influenced by inflammation (similar to ferritin and hepcidin); it might be that, despite adjustment for hs-CRP, inflammation has influenced this specific association. Finally, although we have strong evidence from the literature that FGF23 is a mediator in the association of iron deficiency and EPO with mortality, we cannot exclude that an unmeasured cause of mortality or alternative potential mediators have influenced the current results.

In conclusion, we have shown that functional iron deficiency and higher EPO levels are strongly associated with increased all-cause mortality in the general population. Furthermore, we demonstrated that the increased mortality risk in individuals with diminished iron availability or increased levels of EPO seems to be substantially attributable to variation in FGF23 levels. Future studies will be needed to delineate in more detail the underlying mechanism for the currently identified associations.

Supporting information

S1 Checklist. STROBE checklist. STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. (DOCX)

S1 Fig. Flowchart of the included 6,544 subjects of the PREVEND study. PREVEND, Prevention of Renal and Vascular End-Stage Disease. (DOCX)

S1 Table. Correlation matrix between the different iron status parameters. The table shows the Pearson correlation coefficients. ***P < 0.001. (DOCX)

S2 Table. Determinants of FGF23 levels in the general population. Iron parameters have been placed separately in multivariable analyses; all reported coefficients of the other variables are from the multivariable model, including ferritin. FGF23, fibroblast growth factor 23. (DOCX)

Acknowledgments

We thank Marie M. Pierre Louis for her contribution in the measurement of the FGF23 levels in the PREVEND cohort study.

Author Contributions

Conceptualization: Peter Van der Meer, Stephan J. L. Bakker, Martin H. De Borst.

Data curation: Michele F. Eisenga, Maarten A. De Jong, Peter Van der Meer, Gerwin Huls, Carlo A. J. M. Gaillard, Stephan J. L. Bakker, Martin H. De Borst.

Formal analysis: Michele F. Eisenga, David E. Leaf, Ilja M. Nolte.

Project administration: Stephan J. L. Bakker, Martin H. De Borst.

Supervision: Carlo A. J. M. Gaillard, Stephan J. L. Bakker, Martin H. De Borst.

Visualization: Michele F. Eisenga, David E. Leaf, Ilja M. Nolte.

Writing – original draft: Michele F. Eisenga.

Writing – review & editing: Maarten A. De Jong, Peter Van der Meer, David E. Leaf, Gerwin Huls, Ilja M. Nolte, Carlo A. J. M. Gaillard, Stephan J. L. Bakker, Martin H. De Borst.

References

- GBD 2015 Disease and Injury Incidence and Prevalence Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016; 388: 1545–1602. https://doi.org/10.1016/S0140-6736(16)31678-6 PMID: 27733282
- Stack AG, Mutwali AI, Nguyen HT, Cronin CJ, Casserly LF, Ferguson J. Transferrin saturation ratio and risk of total and cardiovascular mortality in the general population. QJM. 2014; 107: 623–633. <u>https://</u> doi.org/10.1093/gjmed/hcu045 PMID: 24599805
- Souma N, Isakova T, Lipiszko D, Sacco RL, Elkind MS, DeRosa JT, et al. Fibroblast Growth Factor 23 and Cause-Specific Mortality in the General Population: The Northern Manhattan Study. J Clin Endocrinol Metab. 2016; 101: 3779–3786. https://doi.org/10.1210/jc.2016-2215 PMID: 27501282
- Gutierrez OM, Mannstadt M, Isakova T, Rauh-Hain JA, Tamez H, Shah A, et al. Fibroblast growth factor 23 and mortality among patients undergoing hemodialysis. N Engl J Med. 2008; 359: 584–592. <u>https:// doi.org/10.1056/NEJMoa0706130 PMID: 18687639</u>
- Baia LC, Humalda JK, Vervloet MG, Navis G, Bakker SJ, de Borst MH, et al. Fibroblast growth factor 23 and cardiovascular mortality after kidney transplantation. Clin J Am Soc Nephrol. 2013; 8: 1968–1978. https://doi.org/10.2215/CJN.01880213 PMID: 23929933
- Isakova T, Xie H, Yang W, Xie D, Anderson AH, Scialla J, et al. Fibroblast growth factor 23 and risks of mortality and end-stage renal disease in patients with chronic kidney disease. JAMA. 2011; 305: 2432– 2439. https://doi.org/10.1001/jama.2011.826 PMID: 21673295
- Molnar MZ, Tabak AG, Alam A, Czira ME, Rudas A, Ujszaszi A, et al. Serum erythropoietin level and mortality in kidney transplant recipients. Clin J Am Soc Nephrol. 2011; 6: 2879–2886. <u>https://doi.org/10.2215/CJN.05590611</u> PMID: 21980181
- Sinkeler SJ, Zelle DM, Homan van der Heide JJ, Gans RO, Navis G, Bakker SJ. Endogenous plasma erythropoietin, cardiovascular mortality and all-cause mortality in renal transplant recipients. Am J Transplant. 2012; 12: 485–491. https://doi.org/10.1111/j.1600-6143.2011.03825.x PMID: 22054202
- den Elzen WP, Willems JM, Westendorp RG, de Craen AJ, Blauw GJ, Ferrucci L, et al. Effect of erythropoietin levels on mortality in old age: the Leiden 85-plus Study. CMAJ. 2010; 182: 1953–1958. <u>https://</u> doi.org/10.1503/cmaj.100374 PMID: 21149533
- van der Meer P, Voors AA, Lipsic E, Smilde TD, van Gilst WH, van Veldhuisen DJ. Prognostic value of plasma erythropoietin on mortality in patients with chronic heart failure. J Am Coll Cardiol. 2004; 44: 63– 67. https://doi.org/10.1016/j.jacc.2004.03.052 PMID: 15234408
- Drueke TB, Parfrey PS. Summary of the KDIGO guideline on anemia and comment: reading between the (guide)line(s). Kidney Int. 2012; 82: 952–960. https://doi.org/10.1038/ki.2012.270 PMID: 22854645
- Singh AK, Szczech L, Tang KL, Barnhart H, Sapp S, Wolfson M, et al. Correction of anemia with epoetin alfa in chronic kidney disease. N Engl J Med. 2006; 355: 2085–2098. <u>https://doi.org/10.1056/ NEJMoa065485</u> PMID: 17108343
- Swedberg K, Young JB, Anand IS, Cheng S, Desai AS, Diaz R, et al. Treatment of anemia with darbepoetin alfa in systolic heart failure. N Engl J Med. 2013; 368: 1210–1219. https://doi.org/10.1056/ NEJMoa1214865 PMID: 23473338
- 14. Rabadi S, Udo I, Leaf DE, Waikar S, Christov M. Acute blood loss stimulates fibroblast growth factor 23 production. Am J Physiol Renal Physiol. 2017: ajprenal.00081.2017.
- Toro L, Barrientos V, Leon P, Rojas M, Gonzalez M, Gonzalez-Ibanez A, et al. Erythropoietin induces bone marrow and plasma fibroblast growth factor 23 during acute kidney injury. Kidney Int. 2018; 93: 1131–1141. https://doi.org/10.1016/j.kint.2017.11.018 PMID: 29395333
- 16. Hanudel MR, Eisenga MF, Rappaport M, Chua K, Qiao B, Jung G, et al. Effects of erythropoietin on fibroblast growth factor 23 in mice and humans. Nephrol Dial Transplant. 2018; Jul 10 [epub ahead of print].
- Hillege HL, Janssen WM, Bak AA, Diercks GF, Grobbee DE, Crijns HJ, et al. Microalbuminuria is common, also in a nondiabetic, nonhypertensive population, and an independent indicator of cardiovascular risk factors and cardiovascular morbidity. J Intern Med. 2001; 249: 519–526. PMID: <u>11422658</u>

- Eisenga MF, van Londen M, Leaf DE, Nolte IM, Navis G, Bakker SJL, et al. C-Terminal Fibroblast Growth Factor 23, Iron Deficiency, and Mortality in Renal Transplant Recipients. J Am Soc Nephrol. 2017; 28: 3639–3646. https://doi.org/10.1681/ASN.2016121350 PMID: 28774998
- Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF 3rd, Feldman HI, et al. A new equation to estimate glomerular filtration rate. Ann Intern Med. 2009; 150: 604–612. PMID: 19414839
- Mercadal L, Metzger M, Haymann JP, Thervet E, Boffa JJ, Flamant M, et al. A 3-marker index improves the identification of iron disorders in CKD anaemia. PLoS ONE. 2014; 9: e84144. <u>https://doi.org/10. 1371/journal.pone.0084144</u> PMID: 24586229
- Kroot JJ, Laarakkers CM, Geurts-Moespot AJ, Grebenchtchikov N, Pickkers P, van Ede AE, et al. Immunochemical and mass-spectrometry-based serum hepcidin assays for iron metabolism disorders. Clin Chem. 2010; 56: 1570–1579. https://doi.org/10.1373/clinchem.2010.149187 PMID: 20739637
- Pfeiffer CM, Cook JD, Mei Z, Cogswell ME, Looker AC, Lacher DA. Evaluation of an automated soluble transferrin receptor (sTfR) assay on the Roche Hitachi analyzer and its comparison to two ELISA assays. Clin Chim Acta. 2007; 382: 112–116. <u>https://doi.org/10.1016/j.cca.2007.04.008</u> PMID: 17511979
- Heijboer AC, Levitus M, Vervloet MG, Lips P, ter Wee PM, Dijstelbloem HM, et al. Determination of fibroblast growth factor 23. Ann Clin Biochem. 2009; 46: 338–340. <u>https://doi.org/10.1258/acb.2009.</u> 009066 PMID: 19564163
- Preacher K, Hayes A. SPSS and SAS procedures for estimating indirect effects in simple mediation models. Behav Res Methods Instrum Comput. 2004; 36(4):717–731. PMID: 15641418
- 25. Hayes A. Beyond Baron and Kenny: statistical mediation analysis in the new millennium. Commun Monogr. 2009; 76(4): 408–420.
- White IR, Royston P, Wood AM. Multiple imputation using chained equations: Issues and guidance for practice. Stat Med. 2011; 30: 377–399. https://doi.org/10.1002/sim.4067 PMID: 21225900
- 27. Rubin DB. Multiple Imputation For Nonresponse in Surveys. Wiley: New York, New York. 1987.
- Jankowich M, Elston B, Evans SK, Wu WC, Choudhary G. Relationship of Iron Deficiency and Serum Ferritin Levels with Pulmonary Hypertension: The Jackson Heart Study. PLoS ONE. 2016; 11: e0167987. https://doi.org/10.1371/journal.pone.0167987 PMID: 27973582
- 29. Skikne BS. Serum transferrin receptor. Am J Hematol. 2008; 83: 872–875. https://doi.org/10.1002/ajh. 21279 PMID: 18821709
- Goodnough LT. Iron deficiency syndromes and iron-restricted erythropoiesis (CME). Transfusion. 2012; 52: 1584–1592. https://doi.org/10.1111/j.1537-2995.2011.03495.x PMID: 22211566
- Drueke TB, Locatelli F, Clyne N, Eckardt KU, Macdougall IC, Tsakiris D, et al. Normalization of hemoglobin level in patients with chronic kidney disease and anemia. N Engl J Med. 2006; 355: 2071–2084. https://doi.org/10.1056/NEJMoa062276 PMID: 17108342
- 32. Pfeffer MA, Burdmann EA, Chen CY, Cooper ME, de Zeeuw D, Eckardt KU, et al. A trial of darbepoetin alfa in type 2 diabetes and chronic kidney disease. N Engl J Med. 2009; 361: 2019–2032. <u>https://doi.org/10.1056/NEJMoa0907845</u> PMID: 19880844
- Clinkenbeard EL, Hanudel MR, Stayrook KR, Appaiah HN, Farrow EG, Cass TA, et al. Erythropoietin stimulates murine and human fibroblast growth factor-23, revealing novel roles for bone and bone marrow. Haematologica. 2017; 102: e427–e430. <u>https://doi.org/10.3324/haematol.2017.167882</u> PMID: 28818868
- Wolf M, White KE. Coupling fibroblast growth factor 23 production and cleavage: iron deficiency, rickets, and kidney disease. Curr Opin Nephrol Hypertens. 2014; 23: 411–419. https://doi.org/10.1097/01.mnh. 0000447020.74593.6f PMID: 24867675
- Toro L, Barrientos V, Leon P, Rojas M, Gonzalez M, Gonzalez-Ibanez A, et al. Erythropoietin induces bone marrow and plasma fibroblast growth factor 23 during acute kidney injury. Kidney Int. 2018; 93: 1131–1141. https://doi.org/10.1016/j.kint.2017.11.018 PMID: 29395333
- Hanudel MR, Chua K, Rappaport M, Gabayan V, Valore E, Goltzman D, et al. Effects of Dietary Iron Intake and Chronic Kidney Disease on Fibroblast Growth Factor 23 Metabolism in Wild Type and Hepcidin Knockout Mice. Am J Physiol Renal Physiol. 2016: ajprenal.00281.2016.
- Farrow EG, Yu X, Summers LJ, Davis SI, Fleet JC, Allen MR, et al. Iron deficiency drives an autosomal dominant hypophosphatemic rickets (ADHR) phenotype in fibroblast growth factor-23 (Fgf23) knock-in mice. Proc Natl Acad Sci U S A. 2011; 108: E1146–55. <u>https://doi.org/10.1073/pnas.1110905108</u> PMID: 22006328
- McMahon S, Grondin F, McDonald PP, Richard DE, Dubois CM. Hypoxia-enhanced expression of the proprotein convertase furin is mediated by hypoxia-inducible factor-1: impact on the bioactivation of proproteins. J Biol Chem. 2005; 280: 6561–6569. <u>https://doi.org/10.1074/jbc.M413248200</u> PMID: 15611046

- Gutierrez OM, Januzzi JL, Isakova T, Laliberte K, Smith K, Collerone G, et al. Fibroblast growth factor 23 and left ventricular hypertrophy in chronic kidney disease. Circulation. 2009; 119: 2545–2552. https://doi.org/10.1161/CIRCULATIONAHA.108.844506 PMID: 19414634
- 40. Leaf DE, Christov M, Juppner H, Siew E, Ikizler TA, Bian A, et al. Fibroblast growth factor 23 levels are elevated and associated with severe acute kidney injury and death following cardiac surgery. Kidney Int. 2016; 89: 939–948. https://doi.org/10.1016/j.kint.2015.12.035 PMID: 26924052
- Faul C, Amaral AP, Oskouei B, Hu MC, Sloan A, Isakova T, et al. FGF23 induces left ventricular hypertrophy. J Clin Invest. 2011; 121: 4393–4408. https://doi.org/10.1172/JCI46122 PMID: 21985788
- 42. Silswal N, Touchberry CD, Daniel DR, McCarthy DL, Zhang S, Andresen J, et al. FGF23 directly impairs endothelium-dependent vasorelaxation by increasing superoxide levels and reducing nitric oxide bioavailability. Am J Physiol Endocrinol Metab. 2014; 307: E426–36. <u>https://doi.org/10.1152/ajpendo.</u> 00264.2014 PMID: 25053401
- 43. Smith ER, Tan SJ, Holt SG, Hewitson TD. FGF23 is synthesised locally by renal tubules and activates injury-primed fibroblasts. Sci Rep. 2017; 7: 3345–017-02709-w. https://doi.org/10.1038/s41598-017-02709-w PMID: 28611350
- 44. Singh S, Grabner A, Yanucil C, Schramm K, Czaya B, Krick S, et al. Fibroblast growth factor 23 directly targets hepatocytes to promote inflammation in chronic kidney disease. Kidney Int. 2016; 90: 985–996. https://doi.org/10.1016/j.kint.2016.05.019 PMID: 27457912
- Rossaint J, Oehmichen J, Van Aken H, Reuter S, Pavenstadt HJ, Meersch M, et al. FGF23 signaling impairs neutrophil recruitment and host defense during CKD. J Clin Invest. 2016; 126: 962–974. https:// doi.org/10.1172/JCI83470 PMID: 26878171
- 46. Corn PG, Wang F, McKeehan WL, Navone N. Targeting fibroblast growth factor pathways in prostate cancer. Clin Cancer Res. 2013; 19: 5856–5866. https://doi.org/10.1158/1078-0432.CCR-13-1550 PMID: 24052019
- 47. Turner N, Grose R. Fibroblast growth factor signalling: from development to cancer. Nat Rev Cancer. 2010; 10: 116–129. https://doi.org/10.1038/nrc2780 PMID: 20094046
- Feng S, Wang J, Zhang Y, Creighton CJ, Ittmann M. FGF23 promotes prostate cancer progression. Oncotarget. 2015; 6: 17291–17301. https://doi.org/10.18632/oncotarget.4174 PMID: 26019137
- 49. El-Maouche D, Dumitrescu CE, Andreopoulou P, Gafni RI, Brillante BA, Bhattacharyya N, et al. Stability and degradation of fibroblast growth factor 23 (FGF23): the effect of time and temperature and assay type. Osteoporos Int. 2016; 27: 2345–2353. https://doi.org/10.1007/s00198-016-3543-5 PMID: 26928188